Author's response to reviews

Title: GermanIMPACT: Collaborative treatment of late-life depression in primary care - study protocol of a cluster-randomized controlled trial

Authors:

Iris Wernher (wernher@pdx.edu)
Frederike Bjerregaard (frederike.bjerregaard@uniklinik-freiburg.de)
Iris Tinsel (iris.tinsel@uniklinik-freiburg.de)
Christiane Bleich (c.bleich@uke.uni-hamburg.de)
Sigrid Boczor (s.boczor@uke.uni-hamburg.de)
Thomas Kloppe (t.kloppe@uke.uni-hamburg.de)
Martin Scherer (m.scherer@uke.uni-hamburg.de)
Martin Härter (m.haert@uke.uni-hamburg.de)
Wilhelm Niebling (wilhelm.niebling@uniklinik-freiburg.de)
Hans-Helmut König (h.koenig@uke.uni-hamburg.de)
Michael Hüll (M.Huell@zfp-emmendingen.de)

Version: 4
Date: 7 May 2014

Author's response to reviews:

Dear Sir or Madam,

In this revision of the manuscript I added two authors to the list. Due to personnel changes of the study coordination, again two involved co-authors were not listed.

All authors agree to the changes to the author list and on your request they will give their approval in an email.

Faithfully yours,

Frederike Bjerregaard.